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Project Definition:

The ILC Controls effort is on multiple fronts: 

1) Providing a controls system for the SCRF R&D program at Fermilab. This includes NML, Horizontal Test Stand(s), Vertical Test Stand(s) at 1B1. It also includes the controls system for 3.9GHz cavity testing as many of the components being developed at used at A0 as many of these will be moved/deployed at NML. 
2) Global ILC effort. We will help write the RDR including cost estimates. We will help write the EDR and participate in controls R&D necessary for its completion. We will help participate in instrumentation R&D working on HOM BPM measurements and ATCA digitizers.

3) Cavity Data Management System. 
Project Manager’s Summary:
Highlights include
· Interviews for the controls development position were completed.  An offer should be going out soon.  

· Finished paperwork for Michael Davidsaver to start mid-May.

· Starting to bring Bakul Banerjee onboard to help with project planning

ILC: 

FNAL personnel participated in the ATCA Workshop on April 29th which preceded the Real Time Conference hosted by FNAL.  

Data Management:
During the month of April the committee finalized its requirements document and presented it for review by ILC management. The main point of the review was to obtain agreement that process management should be an important component of the final system. This represents a key difference between the DESY and Pansophy systems. We also requested and obtained approval to spend the money necessary to evaluate the Tecnomatix product from UGS.
NML: 
There is a pending request for general wireless network in the 2nd floor rooms and the experiment hall floor.  Carpeting of the control room will be delayed until new HVAC units are installed.  Relay rack installation continues, cable tray installation begins in May.  Kermit Carlson is the “Cable Czar” for the installation and is managing the rack layout and cable requirements.  Discussions have begun for the cavity slow tuner and coupler actuator motor controls.  A new rep-rate generator pc board was designed for the event clock system.

A0:
A new clock event synchronization scheme was tested with Artur Paytyan’s 3.9GHz system.  The new scheme uses a signal derived from the master oscillator 9MHz and no longer needs the LLRF 250KHz LO frequency.  A motor controller was purchased and tested for the new beamline installation.  An event was added to the PhotoInjector clock system to run the coupler conditioning test stand at 2Hz.
Modifications were made to the driver for Alexei Semanov's digitizer board. Alexei had made some hardware modifications to the board and the driver needed to be modified to support the new features.   The new hardware includes two DACs that can be used to set an offset voltage for each of the input channels, enabling the digitizer to support negative polarity signals (-4 to 0V), bipolar signals (-2 to 2 V) and positive polarity signals (0 to 4 V).   Most of this work was motivated by the desire to support Andy Hocker in diagnosing problems he has been having using higher gradients > 17 MV/m in capture cavity II.
Found and fixed triggering problem with OTR cameras. Seven new digital cameras have been ordered and configured for A0 to replace the analog ones. They appear to be working just fine. Worked with Vic on getting images and other DOOCS properties through Matlab (which consumes most of the machine's memory).
Added Vic's OTR Matlab code to the DOOCS cvs tree and included ups directory and files to make it a product. The product path has been added to the MATLABPATH, making the OTR functions available in Matlab.
Asked to get electron beam current through GPIB to the Matlab code. There are a couple of options: use existing server running on the Macintosh or use a DOOCS server. Started implementing the first option based on the Phasescan code - it is a C++ application that connects to a GPIB server running on the Mac, which is connected via GPIB to the scope. The second option is better for long term, but would require hardware (GPIB card or GPIB/Ethernet box) besides having to connect/disconnect from the Mac every time the current is read. While developing application the NFS area suddenly became unavailable. Later found out that fndaut got blocked from the network, even though it had already been patched.
HTS:
Cable installation from the RF hut to the cave is complete.  Cable termination is in progress.  With the exception of solenoid power supply control, the new 1.3GHz RF interlock system is installed and operational.  The 3.9GHz RF interlock system is assembled and being tested.  A temperature interlock for the CC2 tuner motor windings was added to the motor controller.

Dennis Nicklaus helped set up the klystron interlock crates and software for the HTC and CC2 at Meson.  Other small system admin issues were dealt with.  Small maintenance for Coupler Conditioning area.

Infrastructure:
Jerry Firebaugh has the DESY doocs2epics code running on Linux and has done cagets to read data in.  Jimmy You is working on getting IRMIS to crawl the Meson Cryo windows IOC server PVs.
An EPICS naming convention for the test areas was agreed upon that will facilitate the interoperability of DOOCS systems and EPICS systems.   This was based on the naming convention adopted by the High Intensity Neutrino Source. 
A proposal for controls system product layout (EPICS and DOOCS) has been drafted and comments received. A write-up of the EPICS archiving layout at NML has been completed and has generated a lot of questions. We still need work in this area. 

Progress is being made on SPARC board replacement. Software now needs testing by LLRF personnel. 
Vladimir Sirotenko continued to tune the CTF, MDB and NML epics databases.  He installed the application AlwaysUp on the windows epics IOC servers to run epics IOC as a service.  He also implemented a write logging facility for both CTF/MDB and NML nodes with a log server running on smtfs2 node.

Sharon Lackey attended the EPICS Collaboration meeting at DESY, April 23rd through April 27th.   While there she also spent some time learning about the TINE control system and had a tour of the FLASH tunnel.  

Instrumentation R&D:
Ron Rechenmacher and Charlie Briegel prepared for another ATF BPM trip to KEK.  This work included: preparation for install of 6 Echotek boards at KEK, solving interrupt problem on the Fermilab supplied Timing board, solving glitches in data strobe into MVME5500, and understanding Wiener VME Crate nuances.  They left for KEK at the very end of April. 

ATCA:
Resources Used in March 2007 (FTEs):

	
	AD
	CD
	TD
	PPD
	Total Effort

	ILCTA
	2.53
	1.99
	0.00
	0.39
	4.91

	ILC
	0.79
	0.66
	
	
	1.45

	DM
	0.29
	0.11
	0.30
	
	0.70


The effort listed here is time worked and does not include vacation, sick leave, holidays, etc.

Purchase requisitions/procard during April, 2007:  
Misc. components for new rep-rate generator chassis

Stepping motor controller for PhotoInjector beamline

Milestones:

Meetings held, Reports Given:

Weekly status meetings were held  April 2 and April 16.
New Documents posted to ILC DOCDB: 
410-v1 Disk Layout/Productization Proposal
412-v2 ILC Controls Project Summary for March 2007
413-v1 ILCTA Controls Status April 2007
